Author's response to reviews

Title: Is utility-based quality of life associated with overweight in children? Evidence from the UK WAVES randomised controlled study

Authors:

Emma J Frew (e.frew@bham.ac.uk)
Miranda Pallan (m.j.pallan@bham.ac.uk)
Emma Lancashire (e.r.lancashire@bham.ac.uk)
Karla Hemming (k.hemming@bham.ac.uk)
Peymane Adab (p.adab@bham.ac.uk)

Version: 4 Date: 24 November 2015

Author's response to reviews: see over
Dear Editor

Re: MS: 9619021961708166 – Is utility-based quality of life associated with overweight in children? Evidence from the UK WAVES randomised controlled study

Thank you for the opportunity to make further revisions to our paper for publication in BMC Pediatrics. Please find below a point-by-point response to the formatting requests below.

We look forward to hearing from you.

Yours faithfully,

Dr Emma Frew

<p>| | |</p>
<table>
<thead>
<tr>
<th></th>
<th></th>
</tr>
</thead>
<tbody>
<tr>
<td>1. Please include the tables at the end of the manuscript.</td>
<td>Tables now added at end of manuscript.</td>
</tr>
<tr>
<td>2. Please include the figure legend after the References.</td>
<td>Figure now inserted after the references.</td>
</tr>
<tr>
<td>3. Please include the date of the trial registration.</td>
<td>Date added in methods section on page 4, line 89.</td>
</tr>
<tr>
<td>4. Please include the name of the ethics committee who approved the study.</td>
<td>Name of ethics committee added to methods section on page 4, line 91-92.</td>
</tr>
<tr>
<td>5. Please clarify in the Methods if consent was informed and if it was written or oral.</td>
<td>The following sentence has been added to the methods section on page 4, line 96-97: ‘Written parental consent was obtained for each study participant through a signed consent form and verbal assent from the children at the point of measurement.’</td>
</tr>
<tr>
<td>6. Please clarify in the Authors contributions if all authors read and approved the final version of the manuscript.</td>
<td>Statement added to Authors Contributions.</td>
</tr>
</tbody>
</table>